Author's response to reviews

Title: Case report: Perforation of the neovagina in a male to female transsexual

Authors:

Yuto Shimamura (yutshi@luke.ac.jp)
Aoi Fujikawa (aolfjkw@luke.ac.jp)
Keisuke Kubota (kubotake@luke.ac.jp)
Naoki Ishii (naoishi@luke.ac.jp)
Yoshiyuki Fujita (fujiyuki@luke.ac.jp)
Keiichiro Ohta (keiohta@luke.ac.jp)

Version: 3
Date: 18 September 2014

Author's response to reviews: see over
Dear Editor-in-chief, *Journal of Medical Case Reports*

We hereby submit the manuscript entitled “Perforation of the neovagina in a male to female transsexual” to be considered for publication in *Journal of Medical Case Reports* as we have been recommended by editors in *World Journal of Emergency Surgery*.

We experienced a very rare case of acute peritonitis due to perforation of the neovagina in a transsexual. A computed tomography scan image was striking which showed massive intra-abdominal abscess which occupied a significant portion of the intra-abdominal cavity. The perforation of the vaginal vault was confirmed intraoperatively. This case emphasizes the importance of history taking in all details.

We declare that this manuscript is original, has not been reported before and is not currently being considered for publication elsewhere.

We hope you find our manuscript suitable for publication and look forward to hearing from you.

Sincerely,
Yuto Shimamura, M.D.
Division of Gastroenterology
St. Luke’s International Hospital, Tokyo, Japan